Author's response to reviews

Title: Statistical design of personalized medicine interventions: The Clarification of Optimal Anticoagulation through Genetics (COAG) trial

Authors:

Benjamin French (bcfrench@upenn.edu)
Jungnam Joo (jooj@nhlbi.nih.gov)
Nancy L Geller (gellern@nhlbi.nih.gov)
Stephen E Kimmel (stevek@upenn.edu)
Yves Rosenberg (rosenbey@nhlbi.nih.gov)
Jeffrey L Anderson (jeffrey.anderson@imail.org)
Brian F Gage (bgage@dom.wustl.edu)
Julie A Johnson (johnson@cop.ufl.edu)
Jonas H Ellenberg (jellenbe@upenn.edu)

Version: 2 Date: 3 August 2010

Author's response to reviews:

We have included a statement of human subjects approval and consent.